Author's response to reviews

Title: Internet-delivered treatment: Its potential as a low-intensity community intervention for adults with symptoms of depression: Protocol for a randomized controlled trial.

Authors:

Derek Richards (derek.richards@silvercloudhealth.com)
Ladislav Timulak (timulakl@tcd.ie)
Gavin Doherty (gavin.doherty@tcd.ie)
John Sharry (john.sharry@gmail.com)
Amy Colla (amy.colla@silvercloudhealth.com)
Ciara Joyce (ciarajoyce@gmail.com)
Claire Hayes (Claire.Hayes@aware.ie)

Version: 3
Date: 20 December 2013

Author's response to reviews: see over
Editor Trials,

I am enclosing a revised article, a study protocol, entitled “Internet-delivered treatment: Its potential as a low-intensity community intervention for adults with symptoms of depression: Protocol for a randomized controlled trial.”

We have revised the ethics to make it clear and read:

“The study protocol, information on the study, informed consent and related materials have been submitted and approved by the ethics committee at the School of Psychology, Trinity College Dublin (22/11/2013).”

I (Derek Richards) will serve as corresponding author for this manuscript, and assume responsibility for keeping my co-author informed of progress on the submission.

Yours sincerely,

Dr. Derek Richards, University of Dublin, Trinity College, Dublin, Ireland
derek.richards@tcd.ie